Author’s response to reviews

Title: Continuing Medical Education Challenges in Chronic Fatigue Syndrome

Authors:

Dana J Brimmer (dyv4@cdc.gov)
K K McCleary (KKMcCleary@cfids.org)
Teresa A Lupton (tlupton@suddenlink.net)
Katherine M Faryna (yna@carolina.rr.com)
William C Reeves (wcr1@cdc.gov)

Version: 3 Date: 4 November 2009

Author’s response to reviews:

We thank the reviewer for the following suggestion to improve the manuscript titled “Continuing Medical Education Challenges in Chronic Fatigue Syndrome.”

“Please give in Results section heading Conferences-descriptive statistics, only descriptive statistics, and preferably under another heading analyses (on page 7 paragraph 4, heading descriptive statistics for example, it is stated that “Further analysis … showed that conference type … impacted booth visit rates”. Similar statements can be found on page 8 as well.)”

We have changed the headings in the Results section to more accurately reflect the content of this section. The new headings are now,

“Conferences – descriptive statistics”
“Descriptive analysis”
“Logistic regression”

We have revised the manuscript and it now conforms with the BMC Medical Education journal style.